Author's response to reviews

Title: Record linked retrospective cohort study of 4.6 million people exploring ethnic variations in disease: myocardial infarction in South Asians

Authors:

C M Fischbacher (Colin.Fischbacher@isd.csa.scot.nh)
R Bhopal (Raj.Bhopal@ed.ac.uk)
C Povey (Chris.Povey@isd.csa.scot.nhs.uk)
M Steiner (m.steiner@abdn.ac.uk)
J Chalmers (jim.chalmers@isd.csa.scot.nhs.uk)
G Mueller (Ganka.Mueller@gro-scotland.gsi.gov.uk)
J Jamieson (Joan.Jamieson@isd.csa.scot.nhs.uk)
D Knowles (david.knowles@isd.csa.scot.nhs.uk)

Version: 5 Date: 22 June 2007

Author's response to reviews:

dear editor, I believe we have made every one of your required changes, which were mainly formatting. I have also taken the opportunity of updating one of the references in press (no number 31), and correcting typographical and similar errors.

Yours sincerely, Raj Bhopal